# Using patient experience data to support improvements in inpatient mental health care: the EURIPIDES multimethod study

Scott Weich,<sup>1,2\*</sup> Sarah-Jane Fenton,<sup>1,3</sup>
Sophie Staniszewska,<sup>4</sup> Alastair Canaway,<sup>1</sup>
David Crepaz-Keay,<sup>5</sup> Michael Larkin,<sup>6</sup> Jason Madan,<sup>1</sup>
Carole Mockford,<sup>1</sup> Kamaldeep Bhui,<sup>7</sup>
Elizabeth Newton,<sup>8</sup> Charlotte Croft,<sup>9</sup> Una Foye,<sup>7,10</sup>
Aimee Cairns,<sup>1</sup> Emma Ormerod,<sup>11</sup>
Stephen Jeffreys<sup>5,11</sup> and Frances Griffiths<sup>1</sup>

**Declared competing interests of authors:** Scott Weich was a member of the Health Technology Assessment (HTA) Mental, Psychological and Occupational Health Panel (2009–15), the HTA Prioritisation Group (2009–15) and the HTA Clinical Evaluation and Trials Committee (2015–19). Sophie Staniszewska was an associate member of the Health Services and Delivery Research (HSDR) Commissioning Board (researcher led) (2009–18). Carole Mockford was an associate member of the HSDR Commissioning Board (researcher led) (2009–18). Frances Griffiths reports receiving grants from the National Institute for Health Research during the conduct of the study.

Published April 2020 DOI: 10.3310/hsdr08210

<sup>&</sup>lt;sup>1</sup>Warwick Medical School, University of Warwick, Coventry, UK

<sup>&</sup>lt;sup>2</sup>School of Health and Related Research (ScHARR), University of Sheffield, Sheffield, UK

<sup>&</sup>lt;sup>3</sup>Institute for Mental Health, University of Birmingham, Birmingham, UK

<sup>&</sup>lt;sup>4</sup>Warwick Research in Nursing, Warwick Medical School, University of Warwick, Coventry, UK

<sup>&</sup>lt;sup>5</sup>Mental Health Foundation, London, UK

<sup>&</sup>lt;sup>6</sup>School of Life and Health Sciences, Aston University, Birmingham, UK

<sup>&</sup>lt;sup>7</sup>Centre for Psychiatry, Wolfson Institute of Preventative Medicine – Barts and The London, Queen Mary University of London, UK

<sup>&</sup>lt;sup>8</sup>School of Psychology, University of Birmingham, Birmingham, UK

<sup>&</sup>lt;sup>9</sup>Warwick Business School, University of Warwick, Coventry, UK

<sup>&</sup>lt;sup>10</sup>Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

<sup>&</sup>lt;sup>11</sup>National Survivor User Network, London, UK

<sup>\*</sup>Corresponding author s.weich@sheffield.ac.uk

## **Plain English summary**

The EURIPIDES multimethod study

Health Services and Delivery Research 2020; Vol. 8: No. 21

DOI: 10.3310/hsdr08210

NIHR Journals Library www.journalslibrary.nihr.ac.uk

## **Plain English summary**

HS trusts are required to collect patient feedback. We do not know what kinds of feedback are most important or how to translate this feedback into effective action plans to improve services. The Evaluating the Use of Patient Experience Data to Improve the Quality of Inpatient Mental Health Care (EURIPIDES) study looked at how information about patients' experiences is gathered and used in NHS inpatient mental health services in England, to identify how to ensure that this information leads to better services.

We completed five work packages, or substudies, to address five aims. After completing a systematic review of previous studies to identify what matters most to people treated as inpatients (aim 1), we interviewed patient experience leads in NHS mental health trusts in England to ask about approaches to collecting and using patient experience data (aim 2). We selected six trusts with different ways of doing this for an in-depth study and interviewed 182 staff, service users and carers to find out what works best for whom, and where (aim 3). We presented our findings to 44 experts (including experts by experience) at a 'consensus conference' to agree on recommendations about best practice (aim 4). Finally, we used health economics to consider the costs that would arise (and savings that might occur) if best practice in collecting and using patient experience data was widely adopted (aim 5).

Although all mental health trusts collect patient feedback, few analyse or use this to improve services. We identified important findings, from asking for patient feedback about their experiences of inpatient care to receiving, analysing and interpreting, and then acting on, these data to change and improve services. We report these in the form of 18 'rules' to guide trusts on how best to do this. We stress the importance of considering patient experience feedback alongside information about outcomes and safety to identify the areas in which services need to improve.

### **Health Services and Delivery Research**

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **HS&DR** programme

The HS&DR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HS&DR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-services-and-delivery-research.htm

#### This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 14/156/20. The contractual start date was in December 2015. The final report began editorial review in December 2018 and was accepted for publication in September 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Weich et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

#### Editor-in-Chief of **Health Services and Delivery Research** and NIHR Journals Library

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

#### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk